Response to reviewers:

Reviewer #2: “The point of view of this study feels interesting. however, the methods of evaluating the cyst fluid viscosity might be vague and be hard to understand. Please explain the methods of evaluating the cyst fluid viscosity in detail. How did the author draw the flow curve of each case?”

A detailed explanation of the method of generating a rheological curve was added to the manuscript.